Author's response to reviews

Title: Successful desensitization with human insulin using an ultra rush protocol in a patient with insulin allergy and hypersensitivity against protamine

Authors:

Claudia Pfohler (hacpfo@uniklinik-saarland.de)
Cornelia SL Muller (c_mueller1977@yahoo.de)
Dirk O Hasselmann (fetzinger@gmx.de)
Wolfgang Tilgen (hawtil@uniklinik-saarland.de)

Version: 2 Date: 9 January 2008

Author's response to reviews: see over
Claudia Pföhler, MD

To
Prof. Dr. Michael KIDD
University of Sydney
Editor-in-Chief
Journal of Medical Case Reports

Submission of a manuscript for Journal of Medical Case Reports

Dear Prof. Kidd,

Please find enclosed a manuscript entitled “Successful desensitization with human insulin using an ultra rush protocol in a patient with insulin allergy and hypersensitivity against protamine” which we would like to submit for consideration for publication as a case report in the Journal of Medical Case Reports. To our knowledge, this is the first report of a patient suffering from type III insulin allergy in which insulin desensitization was successful. We believe that this form of therapy has a very high clinical impact and is of significant interest for treating physicians e.g. endocrinologists and dermatologists all over the world.

Neither the manuscript, nor parts of it have been submitted elsewhere for publication.

Sincerely yours,

Claudia PFÖHLER, MD